\begin{table}%t5
\caption{Global parameters of BH Vir.}\label{tab5}
%\centering
\par
\begin{tabular}{llllrrlrrrr}
\hline \hline
$i^{\circ}$&$q$&$r_{1}$&$r_{2}$&$T_{1}(K)$&$T_{2}(K)$&L$_{1}^{V}$&$R_{1}$($R_{\odot}$)&$M_{1}$($M_{\odot}$)& $M_{2}$($M_{\odot}$)& References \\
\hline
     &1    &0.26 &0.25 &    &    &    &1.16&0.9 &0.9 &Guiricin et~al. (1984) \\
87  &0.98 &0.25 &0.25 &6000&5500&0.58&1.18&1.00&0.98 &Budding \& Zeilik (\cite{Bud87})\\
87.7&1.03 &     &     &    &    &    &1.05&0.87&0.9  & Strassmeier et~al. (1988)\\
87  &1.02 &0.256&0.235&6000&5850&0.61&1.20&1.02&1.043&Zeilik et~al. (\cite{Zei90}) \\
83.4&0.97 &0.26 &0.24 &    &    &0.67&1.17&0.85&0.83 & Zhai et~al. (\cite{Zha90}) \\
87  &     &0.245&0.225&    &    &0.63&    &    &     &Heckert \& Summers (\cite{Hec95})\\
87  &0.903&0.26&0.237&6095&5520&0.65&1.25&1.165&1.052&Popper (\cite{Pop97}) \\
87.4&0.894&0.263&0.235&6158&5607&   &1.25&1.133&1.013&Clement et~al. (1997)\\
86.5&     &0.255&0.235&6100&5800&0.61&    &     &    &Arevalo et~al. (\cite{Are02})\\
87.4&0.97 &0.255&0.232&6000&5500&0.63&    &     &    &Xiang et~al. (\cite{Xia00})\\
87.6&0.891&0.256&0.235&6000 &5850 &0.65&1.225&1.173&1.046& this paper\\
 \hline
\end{tabular}
\end{table}